Author's response to reviews

Title: What makes health impact assessments successful? Factors contributing to effectiveness in Australia and New Zealand

Authors:

Fiona A Haigh (f.haigh@unsw.edu.au)
Elizabeth Harris (e.harris@unsw.edu.au)
Ben Harris-Roxas (b.harris-roxas@unsw.edu.au)
Fran Baum (fran.baum@flinders.edu.au)
Andrew L Dannenberg (adannen@uw.edu)
Mark F Harris (m.f.harris@unsw.edu.au)
Helen Keleher (helen.keleher@monash.edu)
Lynn Kemp (l.kemp@unsw.edu.au)
Richard Morgan (rm@geography.otago.ac.nz)
Harrison Ng Chok (h.ngchok@unsw.edu.au)
Jeff Spickett (J.Spickett@curtin.edu.au)

Version: 4
Date: 15 September 2015

Author's response to reviews:

Dear Editors,

Please find attached our revised manuscript. As requested we have clarified the role of all the authors in the research project and in the development of the manuscript.

Kind regards

Fiona Haigh